Author’s response to reviews

Title: Decreased fracture incidence with traditional Chinese medicine therapy in patients with osteoporosis: A nationwide population-based cohort study

Authors:

Yu-Chi Wang (cuycuy24@gmail.com)

Jen-Huai Chiang (zinvii@gmail.com)

Hsin-Cheng Hsu (d2762@mail.cmuh.org.tw)

Chun-Hao Tsai (ritsai8615@gmail.com)

Version: 3 Date: 06 Nov 2018

Author’s response to reviews:

Dear Professor Anne Menard

Academic Editor, special issue

BMC Complementary and Alternative Medicine

[BCAM-D-18-00199R2]

Title: “Decreased fracture incidence with traditional Chinese medicine therapy in patients with osteoporosis: A nationwide population-based cohort study.” by Wang et al.

We appreciated the Editor and Reviewers’ efforts and patience when reviewing our article. Your comments and those of the Reviewers were highly insightful and enabled us to improve the quality of our document. Our responses to each comment from the Editor and Reviewers are provided in the following pages on a point-by-point basis.

We hope our revised manuscript will be accepted for publication in the journal BMC Complementary and Alternative Medicine.
Best regards

Correspondence should be addressed to Chun-Hao Tsai, School of Medicine, China Medical University and Department of Orthopedics, China Medical University Hospital, Taichung, Taiwan

No.91 Hsueh-Shih Road, Taichung, TAIWAN E-mail: ritsai8615@gmail.com

Tel: 886-4-22052121 ext. 5052